Author's response to reviews

Title: Anal and oral human papillomavirus (HPV) infection in HIV infected subjects in Northern Italy: a longitudinal cohort study among men who have sex with men.

Authors:

Saverio G Parisi (saverio.parisi@unipd.it)
Mario Cruciani (mcruciani@dronet.org)
Renzo Scaggiante (renzo.scaggiante@sanita.padova.it)
Caterina Boldrin (cateboldrin@libero.it)
Samantha Andreis (andreis.samantha@libero.it)
Federico Dal Bello (federico.dalbello@sanita.padova.it)
Silvana Pagni (silvana.pagni@unipd.it)
Andrea Barelli (Andrea.Barelli@ulss12.ve.it)
Andrea Sattin (andreasattin@yahoo.it)
Carlo Mengoli (carlo.mengoli@unipd.it)
Giorgio Palù (giorgio.palu@unipd.it)

Version: 4 Date: 24 May 2011

Author's response to reviews: see over
Date: may 24, 2011

To The Editor-in-Chief

BMC Infectious Diseases

Subject: Resubmission of manuscript entitled “Anal and oral human papillomavirus (HPV) infection in HIV-infected subjects in northern Italy: a longitudinal cohort study among men who have sex with men.” [Manuscript no. MS: 2107252299506738]

Dear Editor:

We have edited the manuscript, according to the “BMC Series Editorial Production Team” suggestions.

The following changes to the format of our manuscript were made:

Major revisions

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
- addressed

References: Please put citation 44 in square brackets.
- addressed

Minor revisions

Line numbers: Please remove these.
- addressed

Tables- please remove the visible vertical lines from your tables
- addressed

Figures: The figure file should not include the title (e.g. Figure 1... etc.) or the figure number.
- addressed

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:
http://www.biomedcentral.com/info/ifora/figures.
- addressed

We hope that the revised version is now suitable for publication in the BMC Infectious Diseases.

Sincerely,

Saverio G. Parisi